Author’s response to reviews

Title: IgA, albumin, eosinopenia as early warning indicators for cytomegalovirus infection in patients with acute ulcerative colitis

Authors:

Hong Yang (hongy72@163.com)
Kaichun Wu (kaicwu@fmmu.edu.cn)
Hongjie Zhang (hjzhang06@163.com)
Qin Owyang (qin.ouyang@163.com)
Yinglei Miao (myldu@sina.com)
Fang Gu (gf0515@sina.com)
Naizhong Hu (hnz1956@163.com)
Kaifang Zou (zkf@medmail.com.cn)
Jianqiu Sheng (jianqiu@126.com)
Jin Li (15071178164@163.com)
Ping Zheng (zhengpingdoctor@126.com)
Yulan Liu (liuyulan@pkuph.edu.cn)
junxia Li (lijxtom@foxmail.com)
Xiaodi Wang (xdwang60@hotmail.com)
Yongdong Wu (wuyongdong@medmail.com.cn)
Yaozong Yuan (yyz28@medmail.com.cn)
Chunxiao Chen (13906523922@126.com)
Yanhua Pang (yanhuapang@yahoo.com)
Meihua Cui (cuimeih@sina.com)
Jia Ming Qian (qianjm@pumch.cn)
Author’s response to reviews:

Dear Editor in Chief,

Thank you very much for your kind letter and the valuable and constructive comments from the editor and reviewers. We have carefully reviewed the comments and have revised the manuscript accordingly. We hope that the revised manuscript is now suitable for publication in Journal of Clinical Lipidology.

Sincerely, Jiaming Qian, M.D.